1 Building Automation Systems for energy and comfort management in green

- 2 buildings: A critical review and future directions
- 3 4
- Guofeng Qiang^a, Shu Tang^a,*, Jianli Hao^a, Luigi Di Sarno^b, Guangdong Wu^c, Shaoqing Ren^a
- 5
- 6 * School of Design, Xi'an Jiaotong-Liverpool University, 215123, China
- 7 ^bSchool of Engineering, Liverpool University, L693BX, UK
- 8 ^c School of Public Management, Chongqing University, 400044, China
- 9 *Corresponding author.
- 10 *E-mail address*: <u>Shu.Tang@xjtlu.edu.cn</u> (S.Tang).
- 11

12 ABSTRACT

13 Building automation system (BAS) applications in green buildings (GB) is an ideal way to 14 improve energy performance and reduce environmental impact. Although GB has been studied in-15 depth in recent decades, to date, BAS-in-GB research is inferior. Therefore, this study explores the 16 nexus between BAS and GB to achieve a holistic understanding of this area based on 141 articles 17 published from 2008 to 2022. This paper systematically illustrates 1) BAS applications in the 18 lifecycles of GB; 2) BAS applications in supporting GB indoor human comfort: thermal comfort, 19 visual comfort, ventilation comfort, and acoustic comfort; 3) the research framework for reducing 20 the energy performance gap in GB; 4) five BAS and GB integration methods for energy efficiency; 21 and 5) limitations, challenges and future research directions of BAS-in-GB domain. The results 22 show that uncertainties, long-term prediction and control, BAS-supported sustainability goals, and 23 privacy and security are the four main challenging research directions. This study is the first to 24 provide essential guidance on integrating BAS with GB to enhance energy and comfort management. 25 Keywords: Building Automation Systems (BAS), Green buildings (GB), Energy efficiency, 26 Occupant comfort, Energy performance gap (EPG), Integration methods, Sustainability

27

28 **1. Introduction**

29 The conception and technological approaches of green buildings (GB) have gotten broad 30 attention due to the growth of environmental issues and the onset of the energy crisis. The 31 architecture, engineering, and construction (AEC) industries account for nearly 40% of the world's 32 energy, which is also responsible for 32% of CO₂ emissions, 50% of raw materials, and 71% of 33 power use [1–3]. Therefore, countries worldwide advocate for developing GB to reduce energy 34 consumption and CO2 emissions. However, because of subpar design and poor energy system 35 management, they frequently fall short of the required energy performance targets during operation 36 [4]. The energy performance gap (EPG) is the difference in energy consumption between predicted 37 and actual operation. Building automation systems (BAS) application in GB can address this issue 38 by regulating energy efficiency and maintaining a comfortable interior environment over the 39 lifecycle of a building [5].

With the increased interest in GB in recent decades, the number of review articles on the subject is expanding. Integrating quantitative and qualitative review approaches can enhance the depth and breadth of understanding and comprehend the present research state and crucial research challenges [6]. According to previous review articles, GB is a high-performance building that may increase the 44 quality of life while maintaining energy efficiency and reducing the influence on the surrounding 45 environment across the life cycle [6–8]. In the literature, the implementation process, assessment 46 methods, and sustainability performance are three major research streams in the GB domain. Digital 47 technologies, such as building information modeling (BIM), the Internet of Things (IoT), and 48 Artificial Intelligence (AI), are frequently used in the GB area. They enable the integrated 49 administration of information and enhance sustainable performance. Various research has examined 50 their applicability in GB, including BIM for green retrofitting [9], BIM for GB certification systems 51 [7], and AI in GB [10]. These review papers help understand the GB body of knowledge. These 52 studies also showed that the lack of automation control would lead to low energy efficiency and 53 hinder the development of smart GB. BAS can fill this gap due to its ability to integrate multiple 54 functions, including automated control, energy management, occupant comfort, security, and safety 55 [11,12].

56 BAS are distributed systems that control and monitor electrical devices in a building, which 57 include Heating, Ventilation, and Air-conditioning (HVAC), lighting, and alarm security systems 58 [13]. BAS consists of three layers based on function: the field layer (interacting with sensors and 59 actuators), the automation layer (device and process), and the management layer (plant level) [14]. 60 BAS has gained popularity because of its balancing ability to improve occupant comfort and reduce 61 EPG during building operations. [15]. As a recent development, BAS plays a vital role in the AEC 62 sector, particularly in intelligent buildings. Existing review papers mainly focus on energy efficiency 63 optimization [16], automation technology upgrading [17], and occupants' behavior modeling and 64 simulation [18]. They also illustrated that BAS has many potential features to support GB 65 development [19,20]. To sum up: 1) allow prediction of energy performance; 2) reduce time and 66 expense; 3) simplify the measurement of performance indicators; 4) enable input of uncertain 67 parameters; 5) provide comprehensive results. However, existing research has limitations on BAS

68 and GB integration. It is still unknown how to address the following research questions:

- 69 RQ1. What is the current research status of BAS applications in GB?
- 70 RQ2. How to manage energy and comfort by integrating BAS with GB?
- 71 RQ3. What challenges and limitations exist in integrating BAS with GB?

72

RQ4. What are the next steps in BAS-GB research?

73 Therefore, this study aims to conduct a comprehensive review of BAS-in-GB to answer these 74 research questions and benefit the development of GB. The research objectives are: (1) to investigate 75 BAS-supported lifecycles and occupant comfort of GB to construct an EPG framework; (2) to 76 identify and discuss BAS-GB integration methods; (3) to summarize research limitations and 77 challenges; (4) to provide recommendations for future research.

The contributions are listed below. (1) To the best of the authors' knowledge, this is the first article that systematically analyzes the BAS-GB integration for energy and comfort management. 2) This study develops a research framework for reducing EPG. The framework can provide valuable references for researchers and managers. 3) The proposed integration methods expand our knowledge of achieving a trade-off between energy efficiency and occupant comfort. 4) The summarized challenges and recommendations can provide essential guidance for future studies.

84 The paper is structured as follows: Section 2 describes the research methodology. Then Section
85 3 introduces the BAS application in GB. Section 4 discusses hybrid approaches for integrating BAS
86 and GB. Section 5 summarizes research challenges and future directions. Last, Section 6 shows the
87 conclusions.

88 2. Research Methodology

89 The study adopted a mixed method combining two approaches, i.e., the bibliometric approach 90 and systematic literature review (SLR), as shown in Fig.1. The research method includes seven 91 stages: (1) identification, (2) screening, (3) eligibility, (4) included, (5) bibliometric analysis, (6) 92 systematic analysis, and (7) synthesis. This study selects the bibliometric approach as a quantitative 93 method to map and visualize dynamic features of the knowledge field and research trends. The SLR 94 is a qualitative method that can effectively summarize research status, limitations, and future 95 directions. The combination of the two approaches has advantages in providing robust, consistent, 96 and impartial findings [21]. In interdisciplinary BAS-in-GB research domains, it can also increase







Fig. 1. Mixed-methods Systematic review procedure.

103

```
104 Table 1 Keywords for systematic literature review.

String
("Green building" OR "Green construction" OR "High performance building" OR "High
```

performance construction" OR "Sustainable building" OR "Sustainable construction" OR "Green

^{100 2.1.} Literature filtration (stages 1-4)

<sup>This study chose a series of terms based on the subjects of BAS-in-GB. Existing research has
shown that these keywords are valid [5,7,10]. Table 1 lists all keywords.</sup>

project" OR "Sustainable project" OR "Green technology" OR "High-performance building" OR "High-performance construction" OR "High performance project" OR "High-performance project" OR "Energy efficient buildings" OR "Energy performance")

AND

("Building management systems" OR "Building energy systems" OR "Building energy management systems" OR "Building automation systems" OR "Building Control Systems" OR "Building automation and control systems" OR "internet of things" OR "Artificial intelligence" OR "automation" OR "Energy management control systems" OR "Energy management systems" OR "Facility management systems" OR "Building Information Systems" OR "Maintenance Management systems" OR "heating, ventilation, and air conditioning systems")

105

106 The central database used in this article is the Web of Science Core Collection because this 107 database provides peer-reviewed indexed publications which are well-regarded by researchers 108 worldwide. Other databases were also incorporated to allow for a comprehensive examination of 109 the BAS-in-GB area, which includes transdisciplinary themes. They are Science Direct, the 110 American Society of Civil Engineers (ASCE) library, Electrical and Electronics Engineers (IEEE) 111 Xplore, and the Association of Computing Machinery (ACM). The query terms also used the keywords such as "Green buildings," "Building automation systems," and "Energy management 112 113 systems."

114 This study followed five filter criteria to acquire high-quality publications: i) articles or review 115 papers, ii) publication year from 2008 to October 2022, iii) journal impact factor ≥ 1.5 , (iv) full text 116 related to BAS and GB, V) elimination of duplication and irrelevance. Moreover, we added some 117 proceeding articles published by ASCE to make the database more comprehensive. Initially, we 118 filtered 319 publications by title, abstract, and keyword screening. Then we exclude most of them 119 (178 publications) based on the filter criteria. Finally, we rigorously reviewed 141 publications that 120 meet all the requirements. Table 2 depicts the distribution of the 141 publications in the 121 journals/database.

122

No.	Journals/database	Total searching	Reviewed paper	Percentage (%)
1	Energy and Buildings	57	34	24.11
2	Sustainability	41	15	10.64
3	Applied Energy	17	14	9.93
4	Renewable and Sustainable Energy	15	12	8.51
	Reviews			
5	Building and Environment	15	9	6.38
6	Journal of Building Engineering	12	9	6.38
7	Automation in Construction	11	8	5.67
8	Energy	10	3	2.13
9	Journal of Cleaner Production	8	5	3.55
10	Journal of Construction Engineering and Management	7	3	2.13
11	Sustainable City and Society	5	5	3.55
12	Journal of Management in	5	1	0.71
13	Advanced Engineering Informatics	3	2	1.42

14	Journal of Computing in Civil Engineering	2	2	1.42
15	IEEE Xplore Digital Library	33	9	6.38
16	ASCE Digital library	64	7	4.96
17	ACM Digital Library	10	1	0.71
18	Science Direct Library	4	2	1.42
	Total	319	141	100

125 2.2. Bibliometric analysis (step 5)

126 2.2.1. Annual publication trends

Fig. 2 presents the annual publishing tendency of 141 peer-reviewed papers. It shows that there are only ten publications from 2008 to 2012. These publications attempted to find how the design automation strategies affect the energy efficiency of GB [22,23]. Then, the number of publications grew gradually from 2013 to 2017. After that, the number of publications considerably increased, indicating that the BAS-in-GB domain has attained ever-increasing attention.

Furthermore, it is essential to note that the frequency of articles published in 2022 is up to
November. Generally, the exponential growth tendency in Fig. 2 is evidence of growing interest in
the BAS-in-GB field. It will continue to grow in the future.

135





Fig. 2. Number of publications from 2008 to October 2022.

139 This research applied keyword co-occurrence analysis to identify the main focuses in the BAS-140 in-GB field. The co-occurrence analysis helps furnish a holistic understanding of investigated areas 141 in a field. It can also visualize the development trends and intertwined correlations of research topics 142 to support a comprehensive literature review. This study employed VOSviewer software to generate 143 co-occurrence maps, as shown in Fig.3. The co-occurrence network consists of 54 nodes with 1084 144 links. Each node represents a keyword. The 54 keywords meet two filter criteria: (1) their frequency 145 of recurrence is more than three, (2) no duplication. The size of nodes increases based on the 146 frequency of their occurrences.

^{138 2.2.2.} Keyword co-occurrence analysis



147

148

Fig. 3. Keywords co-occurrence network.

150 Table 2 presents 25 terms ranked by co-occurrence values. The overall link strength and 151 average year published can show the degree of attention they have gotten each year. Greater link 152 strength indicates that these keywords have more cross-connections with various study themes. High 153 average citations suggest they have an enormous impact on other studies. For example, the 154 keywords "Energy performance," "Consumption," "Design," and "Energy efficiency" are ranked 155 among the top five in the list, indicating their significance in the field of BAS-in-GB. In addition, 156 the terms "Model," "Simulation," and "Thermal comfort" are top 10 of the list. This result 157 demonstrates that many researchers frequently employ modeling and simulation techniques to 158 enhance thermal comfort in the GB. Additionally, the phrase "Building automation" comes in last 159 on the list, indicating that the investigation into the integration of BAS-GB is lacking.

160 2.2.3 Cluster analysis

161 This study conducts cluster analysis to identify research hotspots and reveals the knowledge structure of the BAS-in-GB field. As presented in Fig. 3, the VOSviewer program classifies these 162 nodes (keywords) into five clusters by employing specific colors: red, green, blue, yellow, and 163 164 purple. Specifically, keywords such as "energy efficiency," "intelligent buildings," "green buildings," "building automation," and "comfort" from cluster 1 (red); "energy performance," 165 "simulation," "AI," "data mining," and "prediction" from cluster 2 (green); "model predictive 166 control," "HVAC," "thermal comfort," and "occupant behavior" from cluster 3 (blue); "design," 167 "management," "sustainability," and "climate" from cluster 4 (yellow); "model," "optimization," 168 169 "machine learning," and "building energy performance" from cluster 5 (purple). Generally, the 170 five primary research areas are as follows: (1) BAS application in the whole lifecycle of GB to 171 improve energy efficiency; (2) prediction and control of HVAC systems to improve occupant 172 comfort; (3) energy consumption simulation and prediction to reduce energy performance gap; (4)

GB-BAS integration methods application for sustainability management; (5) trade-off between
 energy performance and human comfort. The five study fields serve as a framework for systematic

- analysis in the following sections.
- 176

177 Table 2 The top 25 keywords in terms of occurrences.

		Total link			Average year
Rank	Keywords	Occurrences	strength	Cluster	published
1	Performance	26	96	2	2019
2	Consumption	21	77	2	2019
3	Design	20	68	4	2018
4	Energy efficiency	20	72	1	2017
5	Energy performance	18	65	2	2019
6	Model	18	60	5	2019
7	Simulation	18	54	2	2018
8	Thermal comfort	15	74	3	2018
9	Optimization	14	54	5	2019
10	Artificial Intelligence	13	59	2	2017
11	Intelligent buildings	12	40	1	2016
12	Management	12	41	4	2017
13	Green buildings	11	36	1	2017
14	Office buildings	11	31	1	2016
15	Prediction	10	38	2	2018
16	Energy consumption	9	30	2	2017
17	HVAC	9	41	3	2018
18	Machine Learning	9	30	5	2021
19	Building Automation	8	18	1	2016
	Building energy				
20	performance	8	29	5	2018
21	Energy management	8	27	1	2019
22	Internet of Things	8	17	1	2019
23	Model Predictive Control	8	37	3	2017
24	Sustainability	8	30	4	2019
25	Automation	7	22	1	2016

179 2.4. Systematic analysis (stage 6)

After bibliometric analysis, this study conducted SLR to offer in-depth insight into the BASin-GB research field. The five beforementioned research domains are the sites of penetration. Afterward, we divided the five clusters into two groups: BAS application in GB and BAS and GB integration methods. Specifically, the first group investigates how BAS supports GB's lifecycles, occupant comfort, and how to reduce the EPG. The second group presents and analyzes five integration methods extensively utilized for attaining sustainable performance. The subsequent two sections provide more details about SLR.

187 **3. BAS in GB Applications**

188 Previous studies showed that even when buildings fulfill the GB assessment standards such as 189 LEED, there is still considerable EPG [24]. Poor lifecycle management and uncertain occupant 190 behavior are the two leading causes of this phenomenon. The BAS can improve energy efficiency 191 throughout the whole GB lifespan, including four main stages: design, construction, operation, and 192 renovations and retrofit [25]. It can also strengthen occupant comfort without increasing energy 193 consumption. Accordingly, it is significant to investigate how to reduce EPG in GB by applying 194 BAS. Section 3.1 presents the BAS application in GB lifecycles. Section 3.2 illustrates how to 195 enhance occupant comfort through BAS. Section 3.3 develops a research framework that 196 summarizes all solutions for minimizing EPG.

197 3.1. BAS application in GB lifecycles

198 3.1.1. Design stage

199 GB design includes external and internal environmental management [26]. In the external 200 environment, BAS can evaluate the most advantageous sites and building facade through simulation 201 tools for energy conservation. The location should be considered foremost during the design phase. 202 Because several geometric elements, such as the local temperature, the land's natural features, and 203 transportation, considerably influence energy performance [23]. For example, longer transportation 204 distances typically result in higher fuel costs and more vehicle exhaust emissions. Sustainable 205 facade design is another prominent application area. In this domain, BAS is beneficial for lowering 206 construction costs, improving indoor air quality, and conserving energy [27].

In the internal environment, BAS is employed for fire safety [28], energy performance analyses [23,26], HVAC system optimization design [26], and demand-side management [29]. Moreover, scholars have focused on striking a balance between anthropocentrism and mechanical design [30]. Likewise, nature-based biomimetic design progress has become a stream incorporating aesthetics and functionality toward obtaining an intelligent green product [31]. In general, BAS application in the design stage is limited to energy consumption simulation. Therefore, future research should include more real-world examples to support the simulation results.

214 *3.1.2. Construction stage*

In the construction stage, environmental sustainability goals have received increased attention. For example, Alencar et al. [32] applied the Value-Focused Thinking methodology (VFT) to find fundamental factors in the construction stage, including electricity use, water recycling, and construction waste management. They demonstrated that durable and recycled materials could reduce resource waste when building envelope structures. Graziani et al. [33] investigated that the bent welded connectors could economize more electricity than traditional ones without sacrificing the structural integrity.

Additionally, the energy simulation approach is essential in improving energy efficiency. Mathews Roy et al. [34] simulated the building energy consumption of three types of cooling systems, i.e., variable refrigerant flow (VRF) system, variable air volume (VAV) system, and evaporative cooling system through eQuest software. They found that the evaporative cooling system consumes less energy than others.

Furthermore, automated techniques applied in construction safety management have received much interest. For instance, Mascaro et al. [35] developed a robotic walking excavator to automate building the wall using irregularly shaped rocks and rubble. The automated excavator can work in remote and extreme environments, making the building process safer and more efficient. In summary, most of the studies focused on the environmental sustainability of the BAS-in-GB domain.

232 The concern about social and economic factors are very few.

233 *3.1.3. Operation stage*

234 Although GBs are designed based on sustainability certifications like BREEAM, LEED, and 235 China's "Green Building Evaluation Standards." The EPG still exists. In some real cases, actual 236 operation energy savings delivered only 28-71% of design expectations [36]. Therefore, high-237 performance energy management by employing BAS tools emerges from current research. 238 Lertlakkhanakul et al. [37] devised a ubiquitous sensor networks framework to construct the 239 semantic location of BAS to minimize building operation costs. Parise et al. [38] established a 240 procedure based on European Standard EN15193 to evaluate the energy demand for lighting more 241 accurately by considering the productive operating time and occupant behavior. Likewise, 242 Miroshnichenko et al. [39] investigated the influence process of ambient conditions on heat transfer 243 through mathematical simulation. The model they developed can anticipate the amount of energy 244 required to guarantee indoor thermal comfort.

Safety management is equally as crucial as energy management in the operational phase. Building deformation monitoring plays a vital role in the occupants' safety domain. The conventional monitoring technique can only manually acquire data at certain intervals, which is tedious and time-consuming. Thus, researchers created a hybrid BAS-IoT method to address this issue [40,41]. The approach can monitor structural deformation and has advantages in real-time data acquisition, reliable data storage, and precise data analysis [42]. Moreover, BAS can assist in asset management to avoid energy waste and hazards such as pump failures, tank leaks, and creepage.

252 Energy efficiency analyses can benefit from BAS as well. Sang et al. [43] developed two 253 evaluation indicators: coefficient of performance (COP) and energy efficiency ratio. AbdelAzim et 254 al. [44] used Analytic Hierarchy Process (AHP) approach to develop weights for energy efficiency 255 rating criteria. They illustrated that the 'Minimum Energy Performance' factor was mandatory for 256 GBs assessment. In addition, Borowski et al. [45] developed a trigeneration heat and electricity 257 monitoring system for a hotel building, which is more energy efficient than traditional systems. 258 Generally, continuous building facilities monitoring and improving energy efficiency through BAS 259 are critical solutions for reducing EPG.

260 3.1.4. Renovations and retrofit stage

261 Green retrofitting has been the subsequent step in dealing with EPG in recent years [46]. The 262 BAS can help prioritize key performance parameters based on project types. Cost analysis, 263 operational energy, and life cycle analysis are frequently used in the renovations and retrofit stage 264 [9]. Rashid et al. [47] retrofitted a low-cost HVAC system incorporating an occupancy detection 265 system to decrease building energy usage. Engelsgaard et al. [48] introduced an intelligent 266 assessment tool, "IBACSA," which is comprehensive and user-friendly. The tool can quantify the 267 specific impacts of upgrading and retrofitting. They demonstrated its validity in auditing various 268 features in a university based on a qualitative-based multi-criteria approach.

Furthermore, low-impact retrofit solutions are particularly prevalent among researchers. Mauri et al. [49] simulated the relationship between energy consumption and the light transmission of windows by TRNSYS software. They illustrated that residential buildings could achieve up to 44% energy savings when they reduce windows' light transmission to nearly 32%. Ibaseta et al. [50] developed a novel architecture for BAS based on the Web of Things paradigm to address the interoperability issues between heterogeneous devices. They effectively verified the pilot through
an innovative retrofit project in a two-story office building. Rahmawati et al. [51] investigated low
Volatile Organic Compound (VOC) paint and LED bulbs applied in high-rise building refurbishment.
They found that the two methods have advantages in improving indoor air quality and saving energy. *3.1.5. Summary of BAS-supported green building lifecycles*

279 BAS can support various requirements in GB during the whole lifecycle, as shown in Fig.4. In 280 the design stage, BAS enables 3D visualization analyses and simulations to assist designers in 281 optimizing the HVAC system and project scheme. In the construction stage, BAS can offer 282 sustainable material efficiency simulation, waste management, and safe construction processes to 283 achieve sustainable construction objectives. In the operation phase, facility managers can monitor 284 massive building equipment utilizing BAS-based micro-sensor networks to optimize energy usage. 285 Moreover, BAS contains anomaly detection functions for building facilities, such as human health 286 monitoring and fault detection. In the retrofit phase, BAS takes advantage of quantifying the specific 287 impacts of upgrading and retrofitting building envelop, facilities, and appliances in GB. Generally, 288



289 290

Fig. 4. BAS-supported GB lifecycles.

291 3.2. BAS-supported occupant comfort of green building

BAS begins with controlling HVAC, lighting, and other devices to reduce the energy consumption of GB. It progresses to automated adjust indoor environment quality (IEQ) parameters to improve occupant's comfort level [5]. These parameters include thermal, visual, ventilation, and acoustic comfort based on GB assessment criteria like LEED, BREEAM, and Green Star.

296 3.2.1. Thermal comfort

297 In GBs, human thermal comfort is a crucial sustainability metric. It includes a series of 298 notoriously complicated parameters, such as indoor temperature, humidity, and mean radiant 299 temperature [52]. Predicted mean vote (PMV) and predicted percentage of dissatisfaction are two 300 primary thermal comfort assessment approaches in existing studies [2]. Kumar and Hancke [53] 301 developed a wireless, smart comfort sensing system to measure them in real-time efficiently. The 302 system has many desirable qualities, including portability, power efficiency, high reliability, and low 303 system cost. Hu et al. [54] developed intelligent thermal comfort neural network to evaluate human 304 thermal sensations, which outperforms canonical models, such as PMV and adaptive model.

Additionally, HVAC systems provide thermal comfort at the expense of consuming large
 amounts of energy [55,56]. Researchers tried to develop high-efficient HVAC systems to reduce
 power consumption and CO2 emission. However, most systems they introduced are costly due to

complex automation control facilities. Another way to overcome the shortcomings is to integrate
 high-performance building envelopes and thermal energy storage systems [57–59].

Intelligent ventilation systems can also improve indoor air quality and provide thermal comfort for building occupants. The systems mainly adopt intelligent feedback techniques and sensor fusion algorithms to detect HVAC systems and occupancy [60,61]. In addition, making the most of natural resources may provide thermal comfort while remaining economically viable. Examples of this include adaptive envelopes [58], hybrid ventilation systems [62], and earth-based construction materials (EBCMs) [63]. Furthermore, simulating the ground heat transfer and considering weather forecast uncertainties can avoid future thermal imbalance in GBs [39,64].

317 3.2.2. Visual comfort

318 BAS creates visual comfort in adaptive building envelopes, lighting systems, and user 319 interfaces. Initially, the intelligent façade can respond to environmental changes in real time. Several 320 customized components, such as blinds, sunshades, and windows, can adapt to nature's light to 321 maximize visual comfort. These components incorporate outdoor solar radiation measurement to 322 assist users in adjusting inside illumination [58].

Moreover, the BAS can control the amount of daylight through an internal Venetian blind system and feedback to the lighting systems. Subsequently, simulation software such as "Grasshopper" and "Honeybee" evaluate power consumption by inputting lighting parameters such as solar radiation, sunshine duration, and electric lighting [65]. They can also calculate daylight distribution and visualize the 3D visual environment [66].

328 Ultimately, BAS support develops detecting algorithms to simulate occupants' interaction with 329 the building. For example, BAS simulation software employs an image processing algorithm to 330 detect occupancy behavior regarding turning on/off the lighting devices for energy savings [38]. 331 The software "Daysim" enables users to simulate interactive lighting systems through active or 332 passive control [66].

333 3.2.3. Ventilation comfort

People spend more than 90% of their time indoors. Thus, the IEQ plays a significant role in occupants' health and comfort [67]. Frequently natural or mechanical ventilation can improve the air exchange rate between indoor and outdoor environments and remove pollutants in the buildings. In hot regions, such as Malaysia and Australia, a single natural ventilation system is hard to chill the air. As a result, these hot areas must install numerous air conditioning units that consume a lot of electricity.

Additionally, BAS can assist in delivering intelligent ventilation control systems to precisely maintain low levels of power usage and appropriate ventilation air rates [68]. BAS can also provide monitoring systems to measure indoor air parameters, e.g., air temperature, air change rates, pollutant transport, and exposure [55]. In recent years, researchers have integrated the monitoring system with IoT and Big Data techniques to accurately control the pollutant gases like SO2, CO, O3, and NO2 [4].

346 3.2.4. Acoustic comfort

Acoustic comfort substantially impacts IEQ, and most research focuses on improving materials'
sound insulation performance [65,69]. However, few studies concentrate on the BAS application
for acoustic analysis [18,27]. The acoustic effect comes from both internal and external sources.
External acoustic includes vehicles, neighbors, and engineering construction, whereas domestic

activities and appliances are sources of internal acoustic [69]. The application of double-window
and double-skin can decrease the effects of external noise. The internal acoustics parameters are
difficult to identify because of diverse human activities [70].

354 Furthermore, it is challenging to gather acoustic comfort data due to complex data files. 355 Meerbeek [65] collected the data through semi-structured interviews by asking occupants whether 356 they felt comfortable. Pauwels et al. [71] introduced a domain ontology integrating an instance 357 model to check whether the actual acoustic is in line with European acoustic regulations. 358 Nevertheless, recent studies have revealed that IoT-based BMS was able to construct a holistic 359 energy efficiency assessment framework, including acoustic comfort indicators [72–74]. Similarly, 360 Gupta et al. [75] introduced the AI-based BMS framework to extract the features of each acoustic 361 signal.

362 3.2.5. Summary of BAS functions for human comfort in green building

363 BAS applications are designed not only for energy savings but also for occupancy comfort. 364 Indoor environment quality parameters, e.g., thermal comfort, visual comfort, ventilation comfort, 365 and acoustic comfort, influence human comfort level. These parameters could be measured by BAS 366 accurately. Fig. 5 depicts the critical BAS-supported measurement indicators. According to the BAS 367 application mentioned above, thermal and ventilation comfort are more concerned than visual and 368 acoustic comfort. Although BAS provides an efficient and effective analysis tool for IEQ assessment, 369 these comfort variables are separate and cannot evaluate human comfort sensations adequately [54]. 370 Moreover, previous studies have yet to treat these variables in an all-inclusive way to reduce 371 interactive influence [76]. Future research could focus on an integrated green BAS that would 372 improve the human comfort level in the room environment.



373

374 375

Fig. 5. BAS-supported IEQ parameters for GB.

376 3.3. BAS-supported research framework to minimize EPG in GB

According to SLR in sections 3.1 and 3.2, BAS plays a crucial role in enhancing GB's energy efficiency and occupant comfort. Although building energy performance has been modeled and simulated using BAS tools in the design stage, deviation exists from uncertainties, e.g., climate conditions and occupant behavior in the operation stage [77]. These uncertainties will increase the

381 risk of additional energy usage and lead to EPG.

In summary, existing research has made substantial efforts to minimize EPG. For example, predict and control energy performance [34,39,56,78,79], anticipate occupant behavior [18,30,80– 82], and reduce energy consumption [12,44,83–85]. **Fig. 6** presents a research framework for minimizing EPG based on SLR findings from sections 3.1 and 3.2. The diagram shows 16 solutions assigned to four building phases. The three generic solutions in the center of the diagram are consistently valid throughout the building lifecycle.

388



389

390

Fig. 6. Research framework for reducing EPG.

391

392 4. BAS and GB integration methods

Energy and comfort management is the key to reducing EPG. Therefore, this section initially discussed how to predict and control energy efficiency. Next, we analyzed how to achieve a tradeoff between energy consumption and occupant comfort in GB. **Fig.7** shows five BAS-GB integration methods widely employed in the reviewed papers. This research will not present other approaches because they cannot fulfill trade-off objectives.



399

Fig. 7. The percentage of five main integration methods in the reviewed articles.

401

402 *4.1. How to predict and control energy efficiency in GB?*

403 Proportional-Integral-Derivative (PID) and Model Predictive Control (MPC) are the two main 404 energy performance prediction and control models for GB to achieve sustainability goals. However, 405 GB is characteristic of unique and dynamic [86]. Uniqueness indicates that each building has an 406 exceptional location, material, setting, and design standards. Various stakeholders' demands result 407 in dynamics such as construction cost, sustainability goals, and energy usage. These requirements 408 are constantly evolving throughout the whole lifecycles of GB. As a result, simulating the energy 409 efficiency of GB is sophisticated that conventional PID controllers cannot manage BAS efficiently 410 [87]. Therefore, this study introduced an MPC approach to address this issue effectively.

411 4.1.1. Model Predictive Control (MPC) Concept

412 MPC is a simulation model in the building industry used for energy efficiency prediction and 413 control. Fig. 8a illustrates MPC with a classic example of driving a vehicle around. To begin with, 414 the model must consider fundamental variables that influence fuel consumption. Some of these 415 factors are the type of automobile, weight, and aerodynamics. It is the same in GB, and we can 416 improve the envelope by using new insulating materials and double-layer windows to minimize 417 energy consumption [88]. In addition, in the car example, interminable unpredictable situations such 418 as traffic lights, traffic jams, and the number of passengers are hard to regulate. For GB, these 419 variables would be weather conditions and occupant behaviors which are predicted but not 420 controlled. Finally, vehicles can make regular trips to be more comfortable or faster based on 421 people's experience. Before the automobile crosses the curve, MPC controllers can calculate various 422 control actions. Then the best suitable action (direction, speed, and distance) will be taken by 423 modeling the automobile trajectory throughout numerous sample intervals. In a nutshell, it is similar 424 to driving while looking ahead. In a building (Fig. 8b), MPC utilizes an intelligent algorithm to 425 consider the energy consumption from the HVAC and lighting systems. It can also consider a 426 forecast of occupancy behavior and climate conditions to achieve optimal energy management [87].



- 428
- 429

431 4.1.2. Application of Model Predictive Control (MPC) method

432 In literature, modeling and optimization techniques are widely used in MPC to improve energy 433 efficiency in GB operation. There are three main optimization models: physics-based (white box), 434 data-driven (black box), and gray box (hybrid) [16].

435 *Physics-based (white box) model:* The physics-based model is also called the white box model. 436 It is established based on underlying principles of physics, such as mass and energy balances [89]. 437 Current research widely applies the model to optimize the HVAC system design. The physics-based 438 model can simulate the energy performance and indoor environment through soft tools: EnergyPlus 439 [90], TRNSYS [49], eQuest [34], or Modelica [91]. Examples of the application include indoor air 440 quality comfort [92], ground heat exchanger [64], prediction of building heating and cooling 441 demands [93], maintaining thermal comfort through hybrid ventilation [94], and occupancy-based 442 building climate control [19].

443 Data-driven (black box) model: Mathematic equations are the basis of the data-driven model. 444 The model optimization needs to collect massive time series data from actual practice [16]. In the 445 model identification process, statistics and machine learning techniques (e.g., artificial neural 446 network (ANN), multiple linear regression (MLR), support vector machine (SVM)) enable 447 input/output variables without prior knowledge. In the GB field, these approaches have been widely 448 used for structural health monitoring [10,16,95], building energy consumption [52,87,89], occupant 449 behavior prediction [61,80,82,96,97], fault detection and real-time diagnostics [75,93,98,99].

450 Gray-box (hybrid) model: Gray-box (hybrid) model is a compromise between physics-based 451 and data-driven models, which is appropriate for accurate and high-speed calculations in MPC. The 452 gray-box model can also benefit deterministic and stochastic frameworks. Researchers can build the 453 model structure using parameter estimation algorithms and expert knowledge [16]. Thus, the gray-454 box model is effective for HVAC systems when adequate training data is lacking. Additionally, 455 several studies employ a resistance and capacitance (RC) network analog to create a gray-box model 456 for achieving energy savings in GB [19,100,101]. Furthermore, GB dynamics are very complex, 457 and most optimization problems are non-linear, leading to non-convex. Therefore, non-linear 458 programming (NLP) and mixed integer linear programming (MILP) formulation are taken into 459 account to improve energy performance [87,89].

4.1.3. Discussion of Model Predictive Control (MPC) method 460

MPC shows its flexibility in BAS process control while considering the GB dynamics. Its integration with BAS has significant potential for minimizing the impacts arising from uncertainties and maximizing energy performance. Another benefit of this method is that it can predict future disturbances and conflicting events, such as inefficient behavior and harsh climate, and then give optimum management [101]. In addition, the hybrid model is a blend of inductive and deductive models. As a result, it is beneficial to handle non-linear, dynamic, and implicit models derived from real-world data [16].

468 There are also some limitations in MPC: Initially, it needs a large amount of effort to build a 469 suitable model, which increases the investment. The current study indicated that designing and 470 calibrating an adequate model is the most costly activity, accounting for around 70% of the overall 471 project expenses [102]. Next, it presents high requirements for energy managers who need more 472 expertise. Therefore, they need an additional training schema to operate and maintain the MPC. 473 Moreover, the model requires massive data, including weather conditions and occupant behavior 474 which are difficult to be measured in GB. Last, it requires additional computation capacities of 475 hardware.

476 4.2. How to achieve a trade-off between energy efficiency and occupant comfort?

477 Another general objective of BAS is to improve occupant comfort while reducing energy 478 consumption which is inconsistent with the social sustainability goal in GB [103]. Uncertainties present in GB primarily originate from energy system operation, external environment, and 479 480 occupant activity. The uncertainty typically leads to low energy efficiency and makes it difficult to 481 achieve the trade-off goal. Additionally, occupant behavior is a critical determinant of building 482 energy usage. Consequently, current studies have developed several modeling and simulation 483 approaches to address uncertainty elements in BAS. This section will investigate four main 484 integrated methods: agent-based modeling (ABM), data mining (DM), multi-criteria decision-485 making (MCDM), and multi-objective optimization (MOO).

486 *4.2.1. Agent-based modeling*

Fig. 9 shows the classical framework of Agent-based Modeling (ABM), consisting of inputs (building and occupancy) and outputs (comfort) modules. In this framework, local agents represent building devices and control thermal, visual, ventilation, and acoustic comfort [104]. Central agents are responsible for optimizing and controlling the energy performance of the GB through an optimization algorithm. It can communicate with occupants through a graphical user interface and comfortably maintain indoor environmental parameters [15].

493





- 49:
- 496

Fig. 9. Classical framework of Agent-based Modeling (ABM).

497 The ABM integration with BAS can prompt energy saving, optimization, and user comfort. 498 The ABM is characteristic of autonomy, flexibility, homogeneity, and rule-based [17]. It enables 499 simulating the internal and external interaction and evaluating the effects on the BAS. In addition, 500 ABM is the basis of applying artificial intelligence (AI) techniques in BAS, for instance, data mining, 501 ANN, and intelligent decision-making. Furthermore, ABM is crucial to accomplishing multiple 502 goals in innovative energy management. For example, Mousavi et al. [17] developed a new hybrid 503 method by integrating Function Block (software paradigm for industrial automation), intelligent 504 software agents, and semantic web technologies to address BAS complexity arising from 505 heterogeneous components, highly dynamic, and high-level intelligence. The work of Nguyen et al. 506 [105] also revealed that occupant behaviors in buildings could be predicted based on ABM for 507 intelligent energy management.

508 ABMs for occupant behavior investigations have several foci. The majority of studies have 509 focused on how people and building systems interact. For example, Yang and Wang [15] developed 510 a multi-agent-based control framework to address the conflicts between building energy 511 consumption and occupants' comfort. In the beginning, their model divided the whole BAS into 512 several subsystems, helpful in controlling and coordinating multi-agent systems. Then, they created 513 a collection of distributed agents to imitate the occupants' preferences and needs, combining a 514 central agent and a local agent. Finally, they constructed the epistemic, deontic, and axiological 515 (EDA) paradigms. As a result, the agent can make choices based on a thorough investigation of 516 various environmental elements.

517 The ABM is a promising method for integrating various HVAC, lighting, and security systems. 518 It permits the reduction of the disparity between the natural environment values and the user's 519 selected parameters that define the overall comfort level [12]. It can optimize energy management, 520 simulate occupant behavior, and provide intelligent control strategies to reconcile the tension 521 between the human level of satisfaction and power usage, further improving both environment and 522 social sustainability performance in GB.

523 Shortcomings also exist in the present ABM. First, building agents and human agents lack 524 autonomy in many cases. It will lead to only controlled indoor environment sensors by agents with

17

525 authorization. Next, most agents cannot communicate with external environments (software runs 526 out of the BAS). The reasoning technique limits autonomous decisions based on external real-time 527 data. Last, the validation of agent-based simulation, especially stochastic occupancy modeling in a 528 real-world GB project, still needs to be improved.

529 *4.2.2. Data mining*

530 Data mining (DM) is a growing topic in the GB industry that can effectively save time and 531 intelligently identify and analyze energy consumption records available from BAS [106]. It can 532 extract hidden information from semi-structured and unstructured data sources by employing 533 intelligent algorithms. Fig. 10 depictes the DM procedures and applications in GB [107]. This 534 technique has advantages in revealing energy consumption patterns and occupants' preferences 535 compared to traditional statistical models. Artificial neural networks (ANN) and SVM are two 536 common data mining algorithms widely used in GB energy performance systems because of their 537 advantages in addressing complex and non-linear system dynamics [11]. Generally, the DM consists 538 of four steps to analyze energy consumption patterns [106,108], they are: (1) collect data from BAS; 539 (2) pre-process the time-series data and address missing values; (3) analyze the data using the k-540 means clustering method; (4) interpret cluster analysis results.



542

Fig. 10. Framework of DM method in GB.

543

Numerous studies applied DM to discover building energy consumption and saving strategies. 544 545 The research of Chen et al. [96] supported that users would regulate their behavior patterns to 546 decrease energy consumption if provided knowledge about the energy-saving activity. In their work, 547 they used pervasive computing techniques to analyze electricity usage and related activity to find 548 correct energy-efficient behavioral patterns. Then they validated the interactive influence between 549 energy consumption and occupants' behavior using DM algorithms based on intelligent home 550 testbeds. Afroz et al. [16] systematically analyzed the ANN application in the HVAC system, such 551 as the heating/cooling load prediction, the performance of chillers, indoor environment optimization, 552 and dynamics modeling.



In addition, Amayri et al. [82] integrated machine learning and DM techniques to improve the

database quality collected from human behavior in buildings. They adopted a supervised learning approach and Multi-Layer Perceptron (MLP) to construct an occupants' interaction framework. A novel concept, 'spread rate,' was introduced in their research, which can improve the occupancy estimation results and avoid useless interaction with users. Similarly, Alam et al. [106] used an unsupervised data mining approach, 'k-means clustering,' to examine the energy usage trends in educational facilities. They demonstrated that malfunctioning sensors, more extended occupancy, or switch-off mode would affect EPG.

The studies mentioned above explain the interaction patterns of energy consumption and occupant behavior in GB by applying DM techniques. The method possesses several advantages, such as addressing complex non-linear relations between human activity and HVAC systems, discovering hidden energy consumption patterns, and being user-friendly to many researchers. However, several limitations exist. First, no derivation of the produced model is achievable for either DM technique. Next, the performance will become worse when circumstances diverge from training settings. Finally, it is always time-consuming to apply to these programs online.

568 4.2.3. Multi-Criteria Decision Making

569 Multi-Criteria Decision Making (MCDM) is a systematic analysis framework that develops 570 scientific evaluation criteria to solve optimization problems in a group of alternatives [27]. As 571 illustrated in Fig. 11, there are four decision-making processes using MCDM approaches [109]. This 572 new research paradigm is built based on the trade-off objectives of power usage against human 573 comfort. Two conflict indicators are initially selected based on a literature review, workshop, or 574 expert interview. Then, the decision-making matrix is constructed based on the weighing methods, 575 which directly impact the final decision. After that, the alternatives are evaluated and ranked. Finally, 576 designer assesses whether the results meet the trade-off objectives.





Fig.11. MCDM process for achieving a trade-off between energy efficiency and occupant comfort.

580 Researchers have tried to find the most suitable MCDM techniques for intelligent decisions 581 [110]. For instance, Kuo et al. [111] employed fuzzy analytic hierarchical process (FAHP) methods 582 to evaluate the policy efficiency of intelligent GB in Taiwan from 1999 to 2015. Then they collected 583 data through questionnaire surveys and expert interviews. After that, they incorporated a fuzzy 584 transformation matrix (FTM) into FAHP to measure the contributions of each policy. Finally, they 585 verified the weighting results using empirical methods. They found that the effectiveness of policy 586 incentives in the design stage outperforms that in the operation stage. Similarly, Moghtadernejad et 587 al. [27] systematically analyzed nine standard MCDM methods and classified them from two 588 perspectives: problem-solving technique and mathematical nature. The investigation showed that 589 the Choquet Integrals method outperforms other mainly used methods, fuzzy sets, AHP, and 590 TOPSIS, in designing sustainable façades.

591 Furthermore, in the building energy management domain, Starynina and Ustinovichius [112] 592 applied the SyMAD-3 method based on MCDM to determine the most effective modification from 593 composing alternatives. This method could assess the structure parameters of buildings and estimate 594 the energy demand for decision-makers. Notably, the hybrid MCDM methods can calculate energy 595 usage faster and more accurately by reducing the impacts of uncertainty factors generated by human 596 behaviors.

597 Generally, MCDM techniques like fuzzy sets and AHP are easy-to-understand for decision-598 makers. They can effectively assist energy system management which may be difficult when facing 599 conflicting goals between energy consumption and occupant comfort [113]. However, there are also 600 several drawbacks. First, this technique needs to choose reliable energy performance and occupants' 601 comfort criteria based on sufficient experience in the particular field. Moreover, interviewing 602 experts and obtaining a consensus may be difficult because they have various research backgrounds. 603 Last, criteria weighting and ranking results may need to be more consistent when changing the 604 expert panel.

605 4.2.4 Multi-objective optimization (MOO)

606 The multi-objective optimization (MOO) method has been widely applied in GB energy 607 optimization to coordinate conflicting factors and achieve the trade-off objectives before-mentioned 608 [114]. Fig. 12 illustrates four steps included in this optimization method. The inputs are time series data, including energy consumption, weather conditions, and occupant behaviors, extracted from 609 610 the GB modeling software Autodesk series. Then the GB energy model is modeled through 611 simulation programs such as TRNSYS, Energy Plus, and Design Builder. Modeling optimization is 612 followed by optimizing parameters, finding objective functions, and applying optimization 613 algorithms. These algorithms include a Genetic algorithm (GA), Non-dominated Sorting Genetic 614 Algorithm (NSGA-II), and Particle swarm optimization (PSO) [115]. Finally, designers evaluate 615 optimization values to achieve the energy savings and occupants' comfort trade-off objective.



616 617

Fig .12. GB Multi-objective optimization method based on BAS.



619 The MOO method has advantages in controlling and balancing conflicting objectives in the 620 complex BAS [15]. Previous studies have found that it can also support achieving multiple 621 sustainability objectives (social, economic, and environmental) in GB. For example, Lin et al. [116] 622 introduced a MOO method to treat building envelope and BAS in GB. Their work optimizes the 623 design configurations by reducing envelop energy loading and improving air conditioning 624 performance. They found the solution compelling through a case study. The findings showed that 625 the MOO approach decreases CO2 emissions by 58.3% while only increasing costs by 5.3%.

Similarly, to increase the power exchange efficiency based on solar, Middelhauve et al. [117] used the MOO method to identify the most appropriate orientations of photovoltaic panels. Although the design cost rose by 8.3%, the peak power could be reduced by 50% when exchanged with the power grid. Yan [118] argued that adding green criteria to the MOO approach in addition to sustainability and affordability objectives is necessary. He applied the fraction of energy values to calculate their priority efficiently, which can also trade off the three types of optimization objectives.

632 The main advantages of MOO are as follows: (1) It considers energy consumption, weather 633 conditions, and occupants' comfort in parallel and provides valuable data to designers. As a result, 634 it has strengths in identifying the best-compromised solutions by comparing key indices. (2) MOO 635 can solve data integration issues in the BAS, which includes numerous, dynamic, and various energy 636 operation data. However, there are also some limits. (1) Occupant behavior preferences play an 637 essential role in GB energy performance. (2) Existing simulation and modeling software fail to 638 interpolate occupant behavior variables. (3) The present optimization algorithm has yet to adopt 639 holistic optimization methods to address the whole BAS and occupant comfort.

640 4.3. Summary of BAS and GB integration methods

641 In the current research, BAS provides an effective energy performance model and occupant 642 behavior prediction method for GB energy and comfort management. Precisely, to predict and 643 control GB energy consumption, MPC and SMPC are widely employed. These models are 644 characteristics with multi-input multi-output (MIMO), unruled-based, real-time prediction, and 645 control coupled with environmental sustainability assessment in GB. ABM, DM, MCDM, and MOO 646 are the four main methods for modeling energy performance and occupant behavior. These methods 647 cannot only help achieve a trade-off between energy efficiency and occupant comfort but also 648 improve the sustainability level in GB. Fig.13 shows the five integration methods applied for GB 649 trade-off objectives.



653 5. Challenges and recommendations

Although the five main BAS-GB integration methods have many benefits for minimizing EPG
 and achieving trade-off objectives, challenges and limitations also exist. The following sections will
 discuss four main challenges and related research recommendations to provide a valuable reference
 for future work.

658 5.1. Uncertainties affecting energy efficiency

659 Uncertainty is one of the significant challenges for achieving a trade-off between energy 660 consumption and occupant comfort in GB based on BAS. Previous research has shown that occupant 661 behaviors, building operation, and weather conditions are three main uncertainties resulting in GB 662 energy performance discrepancy [87,119]. As shown in Fig.14, building operation status is 663 influenced by weather conditions and occupant behaviors. Moreover, users' behaviors may change 664 in response to weather conditions. For example, occupants will close the window and turn on the 665 air conditioner on a hot day to cool themselves. This behavior will improve thermal comfort and 666 increase electricity consumption at the same time. This causal relationship between them provides 667 a foundation for energy efficiency simulation. Therefore, further research needs to enhance current 668 integration methods to include all three forms of uncertainty.



669 670

671

Fig. 14. Three main uncertainties affecting GB energy efficiency

672 Specifically, occupant behavior is characteristic of highly complex, dynamic, and stochastic in 673 GB [12,82,96]. Researchers employed hybrid ABM, DM, and MPC methods to predict occupant 674 behavior accurately. These studies showed that occupancy or occupant behavior prediction is 675 challenging, and the driving factor is indeterminate [80]. Scholars have developed three 676 fundamental techniques to improve prediction performance: data acquisition, modeling/simulating 677 techniques, and evaluation [18,120,121]. Moreover, future work should create integrated platforms 678 to collect historical data, such as temperature, humidity, and occupants' preference. The platform 679 can construct links with BAS through a central database. It can also support causal analysis of occupant behavior on energy efficiency, help predict and optimize energy usage, and provide realtime BAS management strategies through feedback mechanisms [60,103]. Furthermore, researchers
can adopt a series of software agents as intelligent control devices that provide a friendly human
interface for decision-making [74].

684 Regarding modeling and evaluation techniques, MPC, ABM, DM, MCDM, and MOO 685 implemented in BAS mentioned in section 4 have gained much attention in recent decades. 686 Remarkably, the stochastic MPC framework enables taking all three uncertainties into account based 687 on probability distributions theory and outperforms the deterministic MPC. Although it can provide 688 more reliable energy consumption results [122,123], the three main uncertainty factors mentioned 689 before will also lower its performance [13,101,103]. In addition, researchers should deeply 690 understand integration methods such as ABM, DM, MCDM, and MOO to address uncertainties and 691 achieve GB sustainability. Furthermore, different sensors, devices, and simulation software must 692 work together, leading to interoperability issues. Future studies could apply semantic techniques 693 [124] and develop immersive VR applications based on open-source software tools [95] to address 694 these issues. Last, it is challenging to accurately simulate occupant behavior, weather condition, and 695 building operation status because data acquisition is too tricky. Therefore, further research should 696 acquire more actual and real-time data from residential and commercial buildings to remedy the 697 problem.

698 5.2. Long-term Predictive Control for energy performance

699 Energy consumption in the operation stage of GB is primarily responsible for energy usage 700 throughout the lifecycle. Therefore, it is critical to predict and control energy efficiency from a long-701 term perspective. However, existing forecast methods can only input short-term occupancy data and 702 weather conditions, i.e., hours, weekdays, and months [12,98,125]. Such models can record the 703 occupant behavior in response to indoor environment quality, weather conditions, and adjustments 704 on BAS. However, the strategy is only effective for achieving short-term trade-off goals of energy 705 savings and occupant comfort [64,78]. Future research can combine the regional climate 706 characteristics into their simulation models. For instance, the United States has temperate and 707 subtropical climates, the United Kingdom features a temperate maritime climate, and China is 708 characteristics of hybrid climates [126]. Furthermore, considering human activity events, such as 709 work meetings and academic conferences, can improve the accuracy of long-term predictive 710 analytics of energy performance in GB.

711 During the lifecycle of GB, BAS plays a vital role in energy savings, occupant comfort 712 improvement, and cost reduction, as mentioned in section 3.1. It enables applying intelligent 713 technologies and metering systems to acquire vast amounts of metadata from BAS to monitor and 714 control building facilities [10]. Despite the advantages of BAS in terms of long-term energy 715 performance management, its widespread use in buildings remains a challenge. Future publications 716 can focus on enhancing the interoperability of various platforms, such as BIM, IoT, and DM [127]. 717 BAS integration with them could offer complementary functions and accelerate the digitalization of 718 GB projects. However, most of the research is still at the conceptual stage. Hence, a potential 719 research direction can be integrating BAS with other modeling approaches using advanced 720 technologies, such as semantic web technologies, digital twins, and AI [10,127,128]. Furthermore, 721 future work should consider end-user interactions and feedback in BAS to facilitate increased 722 acceptance [5]. Finally, standardized data acquisition methods are worth investigating to reduce data

723 collection costs.

724 5.3. BAS-supported GB Sustainability Goals

725 GB sustainability goals have three main dimensions, social, economic, and environmental [86]. 726 Existing research mainly concentrates on energy efficiency optimization and occupancy/occupant 727 behavior modeling, which is more related to environmental sustainability [36,39,78]. However, 728 these studies only calculate energy consumption without incorporating the evaluation of CO_2 729 emissions [129]. The lifecycle assessment (LCA) technique can calculate the kg CO_2 emitted 730 accurately and mitigate its impact on the natural environment by providing feedback to design 731 makers [130]. However, the LCA method must be more flexible when design changes occur in GB 732 [131]. Accordingly, the major challenge is integrating this approach with BAS, MCDM [110], and 733 FAHP [132] to assess building energy conservation comprehensively [109]. Therefore, future 734 research should develop an integrated platform to evaluate energy efficiency in GB dynamically.

735 In addition, social and economic sustainability has yet to gain enough attention in the existing 736 BAS-GB domain. A few researchers have attempted to explore stakeholder collaboration for 737 reducing the EPG, which is in line with social sustainability. Seamless knowledge transfer and 738 expertise still need to be improved among stakeholders. Notably, the attention to real-time energy 739 optimization via storage in GB is rising [133,134]. More real-time energy storage and usage data 740 can be acquired from smart grids to promote the development of the energy markets worldwide. This issue may be an effective solution for reducing energy usage costs to achieve economic 741 742 sustainability.

For this reason, the energy storage integrates with BAS in GB will become available at a commercial level in the future. The following are several challenging integration framework developments worthy of future research. (1) The energy transfer framework for cost-effective design [114]; (2) The closed-loop digital twin framework for advanced project management [128]; (3) The building real-time air quality monitoring framework for occupants' health. In conclusion, future work should develop more integrated frameworks to encompass social and economic sustainability.

749 5.4. Privacy and security

750 Tracking human activity through the BAS effectively explores the relationship between energy 751 consumption and occupant behaviors [72]. Smartphone scanning sensors, videos, and cameras can 752 acquire vast volumes of dynamic data in the monitoring process. Operation managers can identify 753 the number of people and their indoor behaviors through these intelligent devices. Indeed, such 754 tracking can provide rich occupant behavior information for high energy efficiency decision-making. 755 However, the approach leads to human activity under real-time supervision, e.g., work, study, and 756 social activities. Therefore, tracking occupants' behavior without invading their privacy is 757 challenging. Another challenge is that large amounts of data need to be recorded and rated for real-758 time processing, even exceeding the capability of smartphones and cameras. Future studies can use 759 heat maps, low-resolution images, and low-quality videos to obtain only vague information about 760 individuals.

Furthermore, current studies construct Personalized Comfort Models (PCMs) to predict and evaluate occupant thermal comfort, incorporating human thermo-regulation factors [135]. Specifically, the model's inputs are physiological signals, e.g., skin temperature, heart rate, and metabolic rate. This model is complementary to the traditional temperature measurement. The application of PCM faces two challenges. The first is that wearable sensors are expensive, especially when testing many people simultaneously [136]. Another is the privacy and security concerns due to tracking physiological signals [124]. Therefore, other thermal comfort measurements, such as air temperature, humidity, and air velocity, can be integrated with BAS as alternatives for future research.

770 6. Conclusions

771 This paper reviews the BAS for energy and comfort management in GB, combing the 772 bibliometric and systematic analysis methods. The quantitative approach analyzes the annual 773 publication trends, keyword co-occurrence, and cluster based on 141 articles in the BAS-GB domain. 774 Additionally, this study systematically analyzes the BAS application in the GB domain and BAS-775 GB integration methods. The results indicate that BAS can minimize EPG and improve occupant 776 comfort throughout the lifecycle of GB. Furthermore, this work discusses five primary BAS-GB 777 integration methods, including MPC, ABM, DM, MCDM, and MOO. Integrating these methods 778 with BAS can achieve a trade-off between energy efficiency and occupant comfort. Last, this review 779 presents challenges and recommendations for future research.

The main contributions of this review are as follows: First, the paper provides theoretical developments for BAS application in GB lifecycles and occupant comfort. It reveals that BAS is a promising tool that can effectively prompt the development of smart GB. Next, this review develops a research framework for reducing EPG in GB. Moreover, this study contributes to developing five integration methods (MPC, ABM, DM, MCDM, and MOO) by combing the BAS technique. These methods will support constructing an integrated platform for achieving a trade-off between low energy consumption and high occupant comfort in GB.

787 Finally, recommendations for future research are proposed based on limitations analysis. They 788 are: (1) improve existing integration methods to contain all three types of uncertainties (occupant 789 behaviors, building operation, and weather conditions); (2) actually simulate the energy 790 consumption in a long-term lifecycle of GB; (3) improve the interoperability of BAS using semantic 791 web technologies; (4) standardized data acquisition methods to reduce data collection cost; (5) 792 construct BAS-GB integration frameworks for achieving social, economic and environmental 793 sustainability goals; (6) develop big data storage techniques and monitor occupants comfort without 794 privacy and security concerns.

795

796 Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

799

800 Acknowledgments

This study is supported by the National Natural Science Foundation of China (NSFC) Young
Scientist Fund (Grant No. 62102324) and Xi'an Jiaotong-Liverpool University Research
Development Fund (Grant No. RDF 20-10-14).

804

805 References

806	[1]	Omrany H, Chang R, Soebarto V, Zhang Y, Ghaffarianhoseini A, Zuo J. A Bibliometric Review
807		of Net Zero Energy Building Research 1995–2022. Energy Build 2022;262:111996.
808		https://doi.org/10.1016/j.enbuild.2022.111996.
809	[2]	Lu Y, Wu Z, Chang R, Li Y. Building Information Modeling (BIM) for green buildings: A
810		critical review and future directions. Autom Constr 2017;83:134-48.
811		https://doi.org/10.1016/j.autcon.2017.08.024.
812	[3]	Inayat A, Raza M. District cooling system via renewable energy sources: A review. Renew
813		Sustain Energy Rev 2019;107:360-73. https://doi.org/10.1016/j.rser.2019.03.023.
814	[4]	Ye Y, Wang Q, Wang J. Green city air monitoring and architectural digital art design based on
815		IoT embedded system. Environ Technol Innov 2021;23:101717.
816		https://doi.org/10.1016/j.eti.2021.101717.
817	[5]	O'Grady T, Chong HY, Morrison GM. A systematic review and meta-analysis of building
818		automation systems. Build Environ 2021;195:107770.
819		https://doi.org/10.1016/j.buildenv.2021.107770.
820	[6]	Wu Z, He Q, Chen Q, Xue H, Li S. A topical network based analysis and visualization of global
821		research trends on green building from 1990 to 2020. J Clean Prod 2021;320:128818.
822		https://doi.org/10.1016/j.jclepro.2021.128818.
823	[7]	Ibrahim O, Imoudu W, Donn M, Chileshe N. Building information modelling and green building
824		certification systems: A systematic literature review and gap spotting. Sustain Cities Soc
825		2022;81:103865. https://doi.org/10.1016/j.scs.2022.103865.
826	[8]	Wang W, Tian Z, Xi W, Tan YR, Deng Y. The influencing factors of China's green building
827		development: An analysis using RBF-WINGS method. Build Environ 2021;188:107425.
828		https://doi.org/10.1016/j.buildenv.2020.107425.
829	[9]	Lim YW, Chong HY, Ling PCH, Tan CS. Greening existing buildings through Building
830		Information Modelling: A review of the recent development. Build Environ 2021:200:107924.
831		https://doi.org/10.1016/j.buildenv.2021.107924.
832	[10]	Debrah C, Chan APC, Darko A. Artificial intelligence in green building. Autom Constr
833		2022;137:104192. https://doi.org/10.1016/j.autcon.2022.104192.
834	[11]	Harish VSKV, Kumar A. A review on modeling and simulation of building energy systems.
835		Renew Sustain Energy Rev 2016;56:1272–92. https://doi.org/10.1016/j.rser.2015.12.040.
836	[12]	Verma A, Prakash S, Kumar A. AI-based Building Management and Information System with
837		Multi-agent Topology for an Energy-efficient Building: Towards Occupants Comfort. IETE J
838		Res 2020. https://doi.org/10.1080/03772063.2020.1847701.
839	[13]	Mirakhorli A, Dong B. Occupancy behavior based model predictive control for building indoor
840		climate—A critical review. Energy Build 2016;129:499–513.
841		https://doi.org/10.1016/j.enbuild.2016.07.036.
842	[14]	Domingues P, Carreira P, Vieira R, Kastner W. Building automation systems: Concepts and
843		technology review. Comput Stand Interfaces 2016;45:1–12.
844		https://doi.org/10.1016/j.csi.2015.11.005.
845	[15]	Yang R, Wang L. Multi-objective optimization for decision-making of energy and comfort
846	_	management in building automation and control. Sustain Cities Soc 2012;2:1-7.
847		https://doi.org/10.1016/j.scs.2011.09.001.
848	[16]	Afroz Z, Shafiullah GM, Urmee T, Higgins G. Modeling techniques used in building HVAC
849	_	control systems: A review. Renew Sustain Energy Rev 2018;83:64-84.

850		https://doi.org/10.1016/j.rser.2017.10.044.
851	[17]	Mousavi A, Vyatkin V. Energy efficient agent function block: A semantic agent approach to IEC
852		61499 function blocks in energy efficient building automation systems. Autom Constr
853		2015;54:127-42. https://doi.org/10.1016/j.autcon.2015.03.007.
854	[18]	Jin Y, Yan D, Chong A, Dong B, An J. Building occupancy forecasting: A systematical and
855		critical review. Energy Build 2021;251:111345. https://doi.org/10.1016/j.enbuild.2021.111345.
856	[19]	Goyal S, Barooah P, Middelkoop T. Experimental study of occupancy-based control of HVAC
857		zones. Appl Energy 2015;140:75-84. https://doi.org/10.1016/j.apenergy.2014.11.064.
858	[20]	Chakraborty D, Elzarka H. Early detection of faults in HVAC systems using an XGBoost model
859		with a dynamic threshold. Energy Build 2019;185:326-44.
860		https://doi.org/10.1016/j.enbuild.2018.12.032.
861	[21]	Snyder H. Literature review as a research methodology: An overview and guidelines. J Bus Res
862		2019;104:333-9. https://doi.org/https://doi.org/10.1016/j.jbusres.2019.07.039.
863	[22]	Caldas L. Generation of energy-efficient architecture solutions applying GENE_ARCH: An
864		evolution-based generative design system. Adv Eng Informatics 2008;22:59-70.
865		https://doi.org/10.1016/j.aei.2007.08.012.
866	[23]	Ochoa CE, Capeluto IG. Strategic decision-making for intelligent buildings: Comparative impact
867		of passive design strategies and active features in a hot climate. Build Environ 2008;43:1829-
868		39. https://doi.org/10.1016/j.buildenv.2007.10.018.
869	[24]	Zou PXW, Alam M. Closing the building energy performance gap through component level
870		analysis and stakeholder collaborations. Energy Build 2020;224:110276.
871		https://doi.org/10.1016/j.enbuild.2020.110276.
872	[25]	Wong JKW, Zhou J. Enhancing environmental sustainability over building life cycles through
873		green BIM: A review. Autom Constr 2015;57:156–65.
874		https://doi.org/10.1016/j.autcon.2015.06.003.
875	[26]	Sha H, Xu P, Yang Z, Chen Y, Tang J. Overview of computational intelligence for building
876		energy system design. Renew Sustain Energy Rev 2019;108:76-90.
877		https://doi.org/10.1016/j.rser.2019.03.018.
878	[27]	Moghtadernejad S, Chouinard LE, Mirza MS. Multi-criteria decision-making methods for
879		preliminary design of sustainable facades. J Build Eng 2018;19:181-90.
880		https://doi.org/10.1016/j.jobe.2018.05.006.
881	[28]	Krause U, Grosshandler W, Gritzo L. The International FORUM of Fire Research Directors: A
882		position paper on sustainability and fire safety. Fire Saf J 2012;49:79-81.
883		https://doi.org/10.1016/j.firesaf.2012.01.003.
884	[29]	Castro M de F, Colclough S, Machado B, Andrade J, Bragança L. European legislation and
885		incentives programmes for demand Side management. Sol Energy 2020;200:114-24.
886		https://doi.org/10.1016/j.solener.2019.12.004.
887	[30]	Hale LA. Anthropocentric urban sustainability: Human significance in building automation.
888		Sustain Cities Soc 2018;42:423-33. https://doi.org/10.1016/j.scs.2018.07.024.
889	[31]	Anghel AA, Giurea D, Mohora I, Hapenciuc A-D, Milincu OC, Frigura-Iliasa FM. Green
890		Interactive Installations as Conceptual Experiments towards a New Meaning of Smart Design.
891		Buildings 2022;12. https://doi.org/10.3390/buildings12010062.
892	[32]	Alencar MH, Priori L, Alencar LH. Structuring objectives based on value-focused thinking
893		methodology: Creating alternatives for sustainability in the built environment. J Clean Prod

- 894 2017;156:62–73. https://doi.org/10.1016/j.jclepro.2017.03.221.
- [33] Graziani L, Quagliarini E, D'Orazio M, Lenci S, Scalbi A. A more sustainableway for producing
 RC sandwich panels on-site and in developing countries. Sustain 2017;9:472.
 https://doi.org/10.3390/su9030472.
- 898 [34] Mathews Roy A, Prasanna Venkatesan R, Shanmugapriya T. Simulation and Analysis of a
 899 Factory Building's Energy Consumption Using eQuest Software. Chem Eng Technol
 900 2021;44:928–33. https://doi.org/10.1002/ceat.202000489.
- 901 [35] Mascaro R, Wermelinger M, Hutter M, Chli M. Towards automating construction tasks: Large902 scale object mapping, segmentation, and manipulation. J F Robot 2021;38:684–99.
 903 https://doi.org/10.1002/rob.22007.
- [36] Zang J, Royapoor M, Acharya K, Jonczyk J, Werner D. Performance gaps of sustainability
 features in green award-winning university buildings. Build Environ 2022;207:108417.
 https://doi.org/https://doi.org/10.1016/j.buildenv.2021.108417.
- 907 [37] Lertlakkhanakul J, Yoon S, Choi J. Developing a building energy management framework based
 908 on ubiquitous sensor networks. Indoor Built Environ 2010;19:192–202.
 909 https://doi.org/10.1177/1420326X09358029.
- 910 [38] Parise G, Martirano L, Parise L. A procedure to estimate the energy requirements for lighting.
 911 IEEE Trans Ind Appl 2016;52:34–41. https://doi.org/10.1109/TIA.2015.2463761.
- 912 [39] Miroshnichenko I V., Sheremet MA, Chen Y Bin, Chang JY. Automation of the heated floor
 913 system in a room under the influence of ambient conditions. Appl Therm Eng 2021;196:117298.
 914 https://doi.org/10.1016/j.applthermaleng.2021.117298.
- 915[40]Tang S, Zhang C, Hao J, Guo F. A Framework for BIM, BAS, and IoT Data Exchange Using916Semantic Web Technologies. Constr Res Congr 2022 Proj Manag Deliv Control Des Mater Sel917PapfromConstrResCongr20222022;3-C:964-73.918https://doi.org/10.1061/9780784483978.098.
- 919[41]Tang S, Shelden DR, Eastman CM, Pishdad-Bozorgi P, Gao X. BIM assisted Building920Automation System information exchange using BACnet and IFC. Autom Constr9212020;110:103049. https://doi.org/10.1016/j.autcon.2019.103049.
- 922[42]Xu Y. Application of Green Building Design Based on the Internet of Things in the Landscape923Planning of Characteristic Towns. Adv Civ Eng 2021;2021.924https://doi.org/10.1155/2021/6317073.
- [43] Sang J, Liu X, Liang C, Feng G, Li Z, Wu X, et al. Differences between design expectations and actual operation of ground source heat pumps for green buildings in the cold region of northern
 (hina. Energy 2022;252:124077. https://doi.org/10.1016/j.energy.2022.124077.
- [44] AbdelAzim AI, Ibrahim AM, Aboul-Zahab EM. Development of an energy efficiency rating
 system for existing buildings using Analytic Hierarchy Process The case of Egypt. Renew
 Sustain Energy Rev 2017;71:414–25. https://doi.org/10.1016/j.rser.2016.12.071.
- [45] Borowski M, Mazur P, Kleszcz S, Zwolińska K. Energy monitoring in a heating and cooling
 system in a building based on the example of the Turówka hotel. Energies 2020;13:1968.
 https://doi.org/10.3390/en13081968.
- [46] Liu C, Ma W, Hao J, Luo D, Zuo J, Zhang C. Energy Retrofitting Assessment of Public Building
 P35 Envelopes in China's Hot Summer and Cold Winter Climate Region. Buildings 2022;12:1866.
 https://doi.org/10.3390/buildings12111866.
- 937 [47] Rashid SA, Haider Z, Chapal Hossain SM, Memon K, Panhwar F, Mbogba MK, et al.

938		Retrofitting low-cost heating ventilation and air-conditioning systems for energy management in
939		buildings. Appl Energy 2019;236:648-61. https://doi.org/10.1016/j.apenergy.2018.12.020.
940	[48]	Engelsgaard S, Alexandersen EK, Dallaire J, Jradi M. IBACSA: An interactive tool for building
941		automation and control systems auditing and smartness evaluation. Build Environ
942		2020;184:107240. https://doi.org/10.1016/j.buildenv.2020.107240.
943	[49]	Mauri L, Vallati A, Ocłoń P. Low impact energy saving strategies for individual heating systems
944		in a modern residential building: A case study in Rome. J Clean Prod 2019;214:791-802.
945		https://doi.org/10.1016/j.jclepro.2018.12.320.
946	[50]	Ibaseta D, García A, Álvarez M, Garzón B, Díez F, Coca P, et al. Monitoring and control of
947		energy consumption in buildings using WoT: A novel approach for smart retrofit. Sustain Cities
948		Soc 2021;65:102637. https://doi.org/10.1016/j.scs.2020.102637.
949	[51]	Rahmawati Y, Utomo C, Sukri NSM, Yasinta RB, Al-Aidrous AHMH. Environmental
950		enhancement through high-rise building refurbishment. Sustain 2020;12:1-14.
951		https://doi.org/10.3390/su12229350.
952	[52]	Čulić A, Nižetić S, Šolić P, Perković T, Čongradac V. Smart monitoring technologies for
953		personal thermal comfort: A review. J Clean Prod 2021;312:127685.
954		https://doi.org/10.1016/j.jclepro.2021.127685.
955	[53]	Kumar A, Hancke GP. An energy-efficient smart comfort sensing system based on the IEEE
956	[]	1451 standard for green buildings. IEEE Sens J 2014;14:4245–52.
957		https://doi.org/10.1109/JSEN.2014.2356651.
958	[54]	Hu W. Wen Y. Guan K. Jin G. Tseng KJ. ITCM: Toward Learning-Based Thermal Comfort
959	[]	Modeling via Pervasive Sensing for Smart Buildings, IEEE Internet Things J 2018:5:4164–77.
960		https://doi.org/10.1109/JIOT.2018.2861831.
961	[55]	Daghigh R. Assessing the thermal comfort and ventilation in Malaysia and the surrounding
962	LJ	regions. Renew Sustain Energy Rev 2015:48:681–91. https://doi.org/10.1016/i.rser.2015.04.017.
963	[56]	Cheng MY, Chiu KC, Lien LC, Wu YW, Lin JJ, Economic and energy consumption analysis of
964	[00]	smart building - MEGA house. Build Environ 2016:100:215–26.
965		https://doi.org/10.1016/i.buildeny.2016.02.023.
966	[57]	Junghans I. Widerin P. Thermal comfort and indoor air quality of the "Concept 22/26" a new
967	[37]	high performance building standard Energy Build 2017:149:114–22
968		https://doi.org/10.1016/i.enhuild.2017.05.020
969	[58]	Shahin HSM Adaptive building envelopes of multistory buildings as an example of high
970	[50]	nerformance building skins Alexandria Eng I 2019:58:345-52
971		https://doi.org/10.1016/i.aei.2018.11.013
072	[50]	Pai CP Surach S Singh VK Bhaysar PP Chandrasakar M Archita V Life cycle assessment
073	[39]	of nanoallow anhanced lawered nerovskite solid solid phase change material till 10000 thermal
973		cycles for energy storage applications. I Energy Storage 2021:35:102220
075		https://doi.org/10.1016/j.est.2020.102220
976	[60]	Lachah F. Bakhauya M. Quladsina P. Essaaidi M. A context driven platform using Internet of
077	[00]	things and data stream processing for heating, ventilation and air conditioning systems control
978		Proce Inst Mech Eng Part I I Syst Control Eng 2010.222.977 99
970		https://doi.org/10.1177/00506518108/153/
080	[61]	Pasalyn ID Ilthra DA Dai A Davarai D Dharadwai D Krishna Kaki SVD Davalagurant and
20U 081		implementation of novel sensor fusion electic for ecourses detection and entered and
701		implementation of novel sensor fusion algorithm for occupancy detection and automation in

982		energy efficient buildings. Sustain Cities Soc 2019;44:85–98.
983		https://doi.org/10.1016/j.scs.2018.09.031.
984	[62]	Rey-Hernández JM, San José-Alonso JF, Velasco-Gómez E, Yousif C, Rey-Martínez FJ.
985		Performance analysis of a hybrid ventilation system in a near zero energy building. Build Environ
986		2020;185. https://doi.org/10.1016/j.buildenv.2020.107265.
987	[63]	Muñoz P, Letelier V, Muñoz L, Zamora D. Assessment of technological performance of extruded
988		earth block by adding bottom biomass ashes. J Build Eng 2021;39:102278.
989		https://doi.org/10.1016/j.jobe.2021.102278.
990	[64]	Dacquay C, Lohrenz E, Fujii H. Ground heat exchanger thermal imbalance prevention using
991		dynamic long-term ground temperature predictions. Int J Green Energy 2021;00:1-9.
992		https://doi.org/10.1080/15435075.2021.1977651.
993	[65]	Meerbeek B, te Kulve M, Gritti T, Aarts M, van Loenen E, Aarts E. Building automation and
994		perceived control: A field study on motorized exterior blinds in Dutch offices. Build Environ
995		2014;79:66-77. https://doi.org/10.1016/j.buildenv.2014.04.023.
996	[66]	Eltaweel A, Yuehong S. Using integrated parametric control to achieve better daylighting
997		uniformity in an office room: A multi-Step comparison study. Energy Build 2017;152:137-48.
998		https://doi.org/https://doi.org/10.1016/j.enbuild.2017.07.033.
999	[67]	Katsoyiannis A, Cincinelli A. 'Cocktails and dreams': the indoor air quality that people are
1000		exposed to while sleeping. Curr Opin Environ Sci Heal 2019;8:6-9.
1001		https://doi.org/https://doi.org/10.1016/j.coesh.2018.12.005.
1002	[68]	Lachhab F, Bakhouya M, Ouladsine R, Essaaidi M. Context-driven monitoring and control of
1003		buildings ventilation systems using big data and Internet of Things-based technologies. Proc Inst
1004		Mech Eng Part I J Syst Control Eng 2019;233:276-88.
1005		https://doi.org/10.1177/0959651818791406.
1006	[69]	Chen J, Qiu Q, Han Y, Lau D. Piezoelectric materials for sustainable building structures:
1007		Fundamentals and applications. Renew Sustain Energy Rev 2019;101:14-25.
1008		https://doi.org/10.1016/j.rser.2018.09.038.
1009	[70]	Benezeth Y, Laurent H, Emile B, Rosenberger C. Towards a sensor for detecting human presence
1010		and characterizing activity. Energy Build 2011;43:305–14.
1011		https://doi.org/10.1016/j.enbuild.2010.09.014.
1012	[71]	Pauwels P, Van Deursen D, Verstraeten R, De Roo J, De Meyer R, Van De Walle R, et al. A
1013		semantic rule checking environment for building performance checking. Autom Constr
1014		2011;20:506-18. https://doi.org/10.1016/j.autcon.2010.11.017.
1015	[72]	Tushar W, Wijerathne N, Li WT, Yuen C, Vincent Poor H, Saha TK, et al. Internet of Things for
1016		Green Building Management: Disruptive Innovations Through Low-Cost Sensor Technology
1017		and Artificial Intelligence. IEEE Signal Process Mag 2018;35:100-10.
1018		https://doi.org/10.1109/MSP.2018.2842096.
1019	[73]	Metwally EA, Farid AA, Ismail MR. Development of an IoT assessment method: An
1020		interdisciplinary framework for energy efficient buildings. Energy Build 2022;254:111545.
1021		https://doi.org/10.1016/j.enbuild.2021.111545.
1022	[74]	Gao X, Pishdad-Bozorgi P, Shelden DR, Tang S. Internet of Things Enabled Data Acquisition
1023		Framework for Smart Building Applications. J Constr Eng Manag 2021;147:04020169.
1024		https://doi.org/10.1061/(asce)co.1943-7862.0001983.
1025	[75]	Gupta N, Khosravy M, Patel N, Dey N, Gupta S, Darbari H, et al. Economic data analytic AI

- 1026technique on IoT edge devices for health monitoring of agriculture machines. Appl Intell10272020;50:3990-4016. https://doi.org/10.1007/s10489-020-01744-x.
- 1028 [76] Salamone F, Barozzi B, Danza L, Ghellere M, Meroni I. Correlation between indoor
 1029 environmental data and biometric parameters for the impact assessment of a living wall in a ZEB
 1030 lab. Sensors (Switzerland) 2020;20:2523. https://doi.org/10.3390/s20092523.
- 1031 [77] Osma G, Amado L, Villamizar R, Ordoñez G. Building Automation Systems as Tool to Improve
 1032 the Resilience from Energy Behavior Approach. Procedia Eng 2015;118:861–8.
 1033 https://doi.org/10.1016/j.proeng.2015.08.524.
- 1034 [78] Kalluri B, Seshadri B, Gwerder M, Schlueter A. A longitudinal analysis of energy consumption
 1035 data from a high-performance building in the tropics. Energy Build 2020;224:110230.
 1036 https://doi.org/10.1016/j.enbuild.2020.110230.
- 1037 [79] Akram MW, Zublie MFM, Hasanuzzaman M, Rahim NA. Global Prospects, Advance
 1038 Technologies and Policies of Energy-Saving and Sustainable Building Systems: A Review.
 1039 Sustain 2022;14:1–27. https://doi.org/10.3390/su14031316.
- I040 [80] Jia M, Srinivasan RS, Raheem AA. From occupancy to occupant behavior: An analytical survey
 I041 of data acquisition technologies, modeling methodologies and simulation coupling mechanisms
 I042 for building energy efficiency. Renew Sustain Energy Rev 2017;68:525–40.
 I043 https://doi.org/10.1016/j.rser.2016.10.011.
- Patti E, Acquaviva A, Jahn M, Pramudianto F, Tomasi R, Rabourdin D, et al. Event-Driven UserCentric Middleware for Energy-Efficient Buildings and Public Spaces. IEEE Syst J
 2016;10:1137–46. https://doi.org/10.1109/JSYST.2014.2302750.
- 1047[82]Amayri M, Ploix S, Bouguila N, Wurtz F. Database quality assessment for interactive learning:1048Applicationtooccupancyestimation.EnergyBuild2020;209.1049https://doi.org/10.1016/j.enbuild.2019.109578.
- 1050 [83] GhaffarianHoseini A, Zhang T, Nwadigo O, GhaffarianHoseini A, Naismith N, Tookey J, et al.
 1051 Application of nD BIM Integrated Knowledge-based Building Management System (BIM1052 IKBMS) for inspecting post-construction energy efficiency. Renew Sustain Energy Rev
 1053 2017;72:935–49. https://doi.org/10.1016/j.rser.2016.12.061.
- 1054 [84] Aste N, Manfren M, Marenzi G. Building Automation and Control Systems and performance
 1055 optimization: A framework for analysis. Renew Sustain Energy Rev 2017;75:313–30.
 1056 https://doi.org/10.1016/j.rser.2016.10.072.
- 1057 [85] Sampaio HV, Westphall CB, Koch F, do Nascimento Boing R, Santa Cruz RN. Autonomic
 1058 energy management with Fog Computing. Comput Electr Eng 2021;93:107246.
 1059 https://doi.org/10.1016/j.compeleceng.2021.107246.
- 1060[86]Zuo J, Zhao ZY. Green building research-current status and future agenda: A review. Renew1061Sustain Energy Rev 2014;30:271-81. https://doi.org/10.1016/j.rser.2013.10.021.
- 1062 [87] Santos-Herrero JM, Lopez-Guede JM, Flores-Abascal I. Modeling, simulation and control tools
 1063 for nZEB: A state-of-the-art review. Renew Sustain Energy Rev 2021;142:110851.
 1064 https://doi.org/10.1016/j.rser.2021.110851.
- 1065 [88] Xing W, Hao J, Ma W, Gong G, Nizami A-S, Song Y. ENERGY PERFORMANCE OF
 1066 BUILDINGS USING ELECTROCHROMIC SMART WINDOWS WITH DIFFERENT
 1067 WINDOW-WALL RATIOS. J Green Build 2022;17:3–20. https://doi.org/10.3992/jgb.17.1.3.
- 1068 [89] Benndorf GA, Wystreil D, Réhault N. Energy performance optimization in buildings: A review
 1069 on semantic interoperability, fault detection, and predictive control. Appl Phys Rev 2018;5.

1070		https://doi.org/10.1063/1.5053110.
1071	[90]	Pinheiro S, Wimmer R, O'Donnell J, Muhic S, Bazjanac V, Maile T, et al. MVD based
1072		information exchange between BIM and building energy performance simulation. Autom Constr
1073		2018;90:91-103. https://doi.org/10.1016/j.autcon.2018.02.009.
1074	[91]	Andriamamonjy A, Saelens D, Klein R. An automated IFC-based workflow for building energy
1075		performance simulation with Modelica. Autom Constr 2018;91:166-81.
1076		https://doi.org/10.1016/j.autcon.2018.03.019.
1077	[92]	Berouine A, Ouladsine R, Bakhouya M, Essaaidi M. Towards a real-time predictive management
1078		approach of indoor air quality in energy-efficient buildings. Energies 2020;13:3246.
1079		https://doi.org/10.3390/en13123246.
1080	[93]	Luo XJ, Oyedele LO, Ajayi AO, Monyei CG, Akinade OO, Akanbi LA. Development of an IoT-
1081		based big data platform for day-ahead prediction of building heating and cooling demands. Adv
1082		Eng Informatics 2019;41:100926. https://doi.org/10.1016/j.aei.2019.100926.
1083	[94]	Chen J, Augenbroe G, Song X. Lighted-weighted model predictive control for hybrid ventilation
1084		operation based on clusters of neural network models. Autom Constr 2018;89:250-65.
1085		https://doi.org/10.1016/j.autcon.2018.02.014.
1086	[95]	Shahinmoghadam M, Natephra W, Motamedi A. BIM- and IoT-based virtual reality tool for real-
1087		time thermal comfort assessment in building enclosures. Build Environ 2021;199:107905.
1088		https://doi.org/10.1016/j.buildenv.2021.107905.
1089	[96]	Chen C, Cook DJ, Crandall AS. The user side of sustainability: Modeling behavior and energy
1090		usage in the home. Pervasive Mob Comput 2013;9:161-75.
1091		https://doi.org/10.1016/j.pmcj.2012.10.004.
1092	[97]	Chen Z, Jiang C. Building occupancy modeling using generative adversarial network. Energy
1093		Build 2018;174:372-9. https://doi.org/10.1016/j.enbuild.2018.06.029.
1094	[98]	Elkhoukhi H, NaitMalek Y, Bakhouya M, Berouine A, Kharbouch A, Lachhab F, et al. A
1095		platform architecture for occupancy detection using stream processing and machine learning
1096		approaches. Concurr Comput Pract Exp 2020;32:1–13. https://doi.org/10.1002/cpe.5651.
1097	[99]	Assaf S, Srour I. Using a data driven neural network approach to forecast building occupant
1098		complaints. Build Environ 2021;200:107972. https://doi.org/10.1016/j.buildenv.2021.107972.
1099	[100]	Zhao Y, Lu Y, Yan C, Wang S. MPC-based optimal scheduling of grid-connected low energy
1100		buildings with thermal energy storages. Energy Build 2015;86:415-26.
1101		https://doi.org/10.1016/j.enbuild.2014.10.019.
1102	[101]	Killian M, Kozek M. Ten questions concerning model predictive control for energy efficient
1103		buildings. Build Environ 2016;105:403-12. https://doi.org/10.1016/j.buildenv.2016.05.034.
1104	[102]	Darivianakis G, Georghiou A, Smith RS, Lygeros J. The Power of Diversity: Data-Driven Robust
1105		Predictive Control for Energy-Efficient Buildings and Districts. IEEE Trans Control Syst
1106		Technol 2019;27:132–45. https://doi.org/10.1109/TCST.2017.2765625.
1107	[103]	Nagpal H, Avramidis II, Capitanescu F, Heiselberg P. Optimal energy management in smart
1108		sustainable buildings – A chance-constrained model predictive control approach. Energy Build
1109		2021;248:111163. https://doi.org/10.1016/j.enbuild.2021.111163.
1110	[104]	Norouziasl S, Jafari A, Wang C. An agent-based simulation of occupancy schedule in office
1111		buildings. Build Environ 2020;186:107352. https://doi.org/10.1016/j.buildenv.2020.107352.
1112	[105]	Nguyen TA, Aiello M. Energy intelligent buildings based on user activity: A survey. Energy
1113		Build 2013;56:244–57. https://doi.org/10.1016/j.enbuild.2012.09.005.

1114	[106]	Alam M. Deviani MP. Analyzing energy consumption patterns of an educational building
1114	[100]	through data mining I Build Eng 2021:44:103385 https://doi.org/10.1016/j.jobe.2021.103385
1115	[107]	Xiao F. Fan C. Data mining in building automation system for improving building operational
1117	[107]	nerformance Energy Build 2014:75:109–18 https://doi.org/10.1016/j.enbuild.2014.02.005
1117	[109]	Ean C. Xiao F. Madeon H. Wang D. Tomporal knowledge discovery in hig BAS date for building
1110	[108]	ran C, Alao F, Madsen H, wang D. Temporal knowledge discovery in org BAS data for building
1120		https://doi.org/10.1016/j.orb/jild.2015.00.060
1120	[100]	nttps://doi.org/10.1016/j.enbuild.2015.09.060.
1121	[109]	Chen L, Pan W. Review fuzzy multi-criteria decision-making in construction management using
1122		a network approach. Appl Soft Comput $2021;102:10/103$.
1123	[110]	https://doi.org/10.1016/j.asoc.2021.10/103.
1124	[110]	Campos-Guzmán V, Garcia-Cáscales MS, Espinosa N, Urbina A. Life Cycle Analysis with
1125		Multi-Criteria Decision Making: A review of approaches for the sustainability evaluation of
1126		renewable energy technologies. Renew Sustain Energy Rev 2019;104:343-66.
1127		https://doi.org/10.1016/j.rser.2019.01.031.
1128	[111]	Kuo CFJ, Lin CH, Hsu MW, Li MH. Evaluation of intelligent green building policies in Taiwan
1129		- Using fuzzy analytic hierarchical process and fuzzy transformation matrix. Energy Build
1130		2017;139:146–59. https://doi.org/10.1016/j.enbuild.2016.12.078.
1131	[112]	Starynina J, Ustinovichius L. A multi-criteria decision-making synthesis method to determine
1132		the most effective option for modernising a public building. Technol Econ Dev Econ
1133		2020;26:1237–62. https://doi.org/10.3846/tede.2020.13398.
1134	[113]	Antucheviciene J, Kala Z, Marzouk M, Vaidogas ER. Solving Civil Engineering Problems by
1135		Means of Fuzzy and Stochastic MCDM Methods: Current State and Future Research. Math Probl
1136		Eng 2015;2015. https://doi.org/10.1155/2015/362579.
1137	[114]	Seghier TE, Lim YW, Harun MF, Ahmad MH, Samah AA, Majid HA. BIM-based retrofit
1138		method (RBIM) for building envelope thermal performance optimization. Energy Build
1139		2022;256:111693. https://doi.org/10.1016/j.enbuild.2021.111693.
1140	[115]	Lu Y, Wang S, Zhao Y, Yan C. Renewable energy system optimization of low/zero energy
1141		buildings using single-objective and multi-objective optimization methods. Energy Build
1142		2015;89:61-75. https://doi.org/10.1016/j.enbuild.2014.12.032.
1143	[116]	Lin YH, Lin M Der, Tsai KT, Deng MJ, Ishii H. Multi-objective optimization design of green
1144		building envelopes and air conditioning systems for energy conservation and CO2 emission
1145		reduction. Sustain Cities Soc 2021;64:102555. https://doi.org/10.1016/j.scs.2020.102555.
1146	[117]	Middelhauve L, Baldi F, Stadler P, Maréchal F. Grid-Aware Layout of Photovoltaic Panels in
1147		Sustainable Building Energy Systems. Front Energy Res 2021;8:1–19.
1148		https://doi.org/10.3389/fenrg.2020.573290.
1149	[118]	Su Y. Smart energy for smart built environment: A review for combined objectives of affordable
1150		sustainable green. Sustain Cities Soc 2020;53:101954. https://doi.org/10.1016/j.scs.2019.101954.
1151	[119]	Lork C, Li WT, Qin Y, Zhou Y, Yuen C, Tushar W, et al. An uncertainty-aware deep
1152		reinforcement learning framework for residential air conditioning energy management. Appl
1153		Energy 2020;276:115426. https://doi.org/10.1016/j.apenergy.2020.115426.
1154	[120]	Pallikere A, Qiu R, Delgoshaei P, Negahban A. Incorporating occupancy data in scheduling
1155		building equipment: A simulation optimization framework. Energy Build 2020;209.
1156		https://doi.org/10.1016/j.enbuild.2019.109655.
1157	[121]	Liang X, Hong T, Shen GQ. Occupancy data analytics and prediction: A case study. Build

Environ 2016;102:179-92. https://doi.org/10.1016/j.buildenv.2016.03.027.
22] Wu C, Gao S, Liu Y, Song TE, Han H. A model predictive control approach in microgrid
considering multi-uncertainty of electric vehicles. Renew Energy 2021;163:1385-96
https://doi.org/10.1016/j.renene.2020.08.137.
Hu J, Shan Y, Guerrero JM, Ioinovici A, Chan KW, Rodriguez J. Model predictive control of
microgrids – An overview. Renew Sustain Energy Rev 2021;136:110422
https://doi.org/10.1016/i.rser.2020.110422.
24] Hu S. Wang J. Hoare C. Li Y. Pauwels P. O'Donnell J. Building energy performance assessment
using linked data and cross-domain semantic reasoning. Autom Constr 2021:124:103580
https://doi.org/10.1016/i.autcon.2021.103580
25] Parker A Moavedi S James K Peng D Alahmad MA A Case Study to Quantify Variability in
Building Load Profiles IFFE Access 2021.9.127799–813
https://doi.org/10.1109/ACCESS 2021.3112103
26] Kitoh A Endo H Krishna Kumar K Cavalcanti IEA Goswami P Zhou T Monsoons in a
changing world: A regional perspective in a global context. I Geophys Res Atmos
2012:118:2052 (5 https://doi.org/10.1002/j.grd.50258
2013,118.3035-03. https://doi.org/10.1002/jgrd.30238.
[27] Tang S, Sheiden DK, Eastman CM, Pishdad-Bozorgi P, Gao X. A review of building information modeling (DDM) and the interret of things (LT) devices integration. Present status and fiture
modeling (BIM) and the internet of things (101) devices integration: Present status and future
trends. Autom Constr 2019;101:127–39. https://doi.org/10.1016/j.autcon.2019.01.020.
[28] Pan Y, Zhang L. A BIM-data mining integrated digital twin framework for advanced project
management. Autom Constr 2021;124:103564. https://doi.org/10.1016/j.autcon.2021.103564.
[29] Hao JL, Cheng B, Lu W, Xu J, Wang J, Bu W, et al. Carbon emission reduction in prefabrication
construction during materialization stage: A BIM-based life-cycle assessment approach. Sci
Total Environ 2020;723:137870. https://doi.org/10.1016/j.scitotenv.2020.137870.
[30] Andler J, Mathur N, Sutherland J, Zhao F, Handwerker C. Guiding the environmental design of
a novel solar absorber through life cycle assessment by identifying anticipated hot spots. J Clear
Prod 2020;258:120847.
31] Zhang N, Zhang D, Zuo J, Miller TR, Duan H, Schiller G. Potential for CO2 mitigation and
economic benefits from accelerated carbonation of construction and demolition waste. Renew
Sustain Energy Rev 2022;169:112920. https://doi.org/10.1016/j.rser.2022.112920.
32] Lazar N, Chithra K. Prioritization of sustainability dimensions and categories for residential
buildings of tropical climate: A multi-criteria decision-making approach. J Build Eng
2021;39:102262. https://doi.org/10.1016/j.jobe.2021.102262.
33] Georgiou GS, Christodoulides P, Kalogirou SA. Real-time energy convex optimization, via
electrical storage, in buildings - A review. Renew Energy 2019;139:1355-65
https://doi.org/10.1016/j.renene.2019.03.003.
[34] Guo Y, Wang G, Xu R, He R, Wei X, Tellambura C. Capacity Analysis for Wireless Symbiotic
Communication Systems with BPSK Tags under Sensitivity Constraint. IEEE Commun Lett
2022;26:44-8. https://doi.org/10.1109/LCOMM.2021.3125342.
35] Li D, Menassa CC, Kamat VR. Personalized human comfort in indoor building environments
under diverse conditioning modes. Build Environ 2017;126:304-17
https://doi.org/10.1016/j.buildenv.2017.10.004.
[36] Lee J, Ham Y. Physiological sensing-driven personal thermal comfort modelling in consideration
of human activity variations. Build Res Inf 2021;49:512–24

1202 https://doi.org/10.1080/09613218.2020.1840328.